Author's response to reviews

Title: The role of 25-hydroxyvitamin D deficiency in promoting insulin resistance and inflammation in patients with Chronic Kidney Disease: a randomised controlled trial

Authors:

William G Petchey (william_petchey@health.qld.gov.au)
Ingrid J Hickman (i.hickman@uq.edu.au)
Emma Duncan (e.duncan@uq.edu.au)
Johannes B Prins (j.prins@uq.edu.au)
Carmel M Hawley (carmel_hawley@health.qld.gov.au)
David W Johnson (david_johnson@health.qld.gov.au)
Katherine Barraclough (katherine_barraclough@health.qld.gov.au)
Nicole M Isbel (nikky_isbel@health.qld.gov.au)

Version: 3 Date: 1 December 2009

Author's response to reviews: see over
Dear Ms Neilan,

Re: manuscript 1560484705297881: The role of 25-hydroxyvitamin D deficiency in promoting insulin resistance and inflammation in patients with Chronic Kidney Disease: a randomised controlled trial.

Thank you for the email dated 1 December 2009. The manuscript has now been revised as per your instructions and has been uploaded on your website. The following changes have been made:

1. Abstract reformatted with inclusion of trial registry information in methods section, and inclusion of discussion section.
2. Abstract reference parentheses have been reformatted to meet guidelines (non-superscript).
3. Revision of text for typographical errors and final corrections

I hope this is all in order, but should there be any further questions or problems with the manuscript please contact me via email or on the numbers above.

With kind regards.

Yours sincerely

Dr Will Petchey
Renal Research Fellow